Author's response to reviews

Title: Inferior vena cava prosthetic replacement in a patient with horseshoe kidney and metastatic testicular tumor: technical considerations and review of the literature

Authors:

Pietro Rispoli (pietro.rispoli@unito.it)
Paolo Destefanis (p.deste@gmail.com)
Paolo Garneri (grn_plo80@hotmail.it)
Gianfranco Varetto (varetto@hotmail.com)
Beatrice Lillaz (blillaz@cittadellasalute.to.it)
Libero Ciuffreda (lciuffreda@cittadellasalute.to.it)
Dario Fontana (dario.fontana@unito.it)

Version: 2 Date: 30 January 2014

1) We corrected the title: now the title in the submission system is consistent with that of the manuscript. We also added email adresses of all authors in the title page as required.

2) We added acknowledgements as required: we didn't need funds for the drafting of the article. Were not involved other people outside of the authors.

3) We added line numbers in the manuscript as required.